Author’s response to reviews

Title: The prevalence and impact of self-reported foot and ankle pain in the over 55 age group: a secondary data analysis from a large community sample

Authors:

Anne-Maree Keenan (a.keenan@leeds.ac.uk)

Chris Drake (Chris.Drake@midyorks.nhs.uk)

Philip Conaghan (p.conaghan@leeds.ac.uk)

Alan Tennant (a.tennant@leeds.ac.uk)

Version: 2 Date: 24 Oct 2019

Author’s response to reviews:

Dear Dr Buldt,

Thank-you for your correspondence and reviewer feedback on our submitted article “The prevalence and impact of self-reported foot and ankle pain in the over 55 age group: data analysis from a large community sample”.

We are apologies for our oversight of Reviewer 1 comments. We have clarified that the questionnaires were posted to participants under the methods section. We apologise for our misunderstanding which meant that we did not include the 95% confidence intervals for our prevalence data. This has now been provided in Table 2.